



Documento de Trabajo/Working Paper Serie Economía

Inequality of opportunity in Europe

by

Gustavo A. Marrero and Juan G. Rodríguez

June 2011

DT-E-2011-08

ISSN: 1989-9440

INEQUALITY OF OPPORTUNITY IN EUROPE

Gustavo A. Marrero

Departamento de Análisis Económico (Universidad de La Laguna, Spain). Tel: +34 922 317123.

E-mail: gmarrero@ull.es

Juan G. Rodríguez

Departamento de Fundamentos del Análisis Económico I (Universidad Complutense de Madrid). Tel: +34 91 3942515.

E-mail: juangabriel.rodriguez@ccee.ucm.es

Abstract

Using the EU-SILC database, we estimate and compare the Inequality of Opportunity (IO) of 23 European countries in 2005. IO is estimated as the between-type (*ex-ante*) inequality component following the parametric procedure of Ferreira and Gignoux (2011), which allows for the inclusion of the large set of circumstances in the database. As a modest attempt to understand some reasons behind IO differences, we also measure the degree of correlation between IO estimates and a set of past and contemporaneous economic factors related to the degree of development, labor market performance, investment in human capital and social protection spending.

JEL Classification: D63, E24, O15, O40.

Key Words: Inequality of Opportunity; EU-SILC database; circumstances; Europe.

1. Introduction

Equality of opportunity has traditionally been understood as the absence of barriers to access positions, education and jobs. In line with this conception, hiring should be meritocratic and characteristics like economic class, gender and race should have no bearing on the merit of the individual (Lucas, 1995). Rawls (1971) and Sen (1980; 1985), among others, invoked a more general notion of equality of opportunity. They argued that equality of opportunity would require compensating persons for a variety of *circumstances* (i.e., socioeconomic background, ethnicity, place of birth, etc.) whose distribution is morally arbitrary.¹

Roemer (1993, 1996 and 1998) brings that philosophical debate into economics and formalizes a precise definition of equality of opportunity.² He emphasizes that an individual's outcome (income, welfare, health, etc.) is a function of variables *within* and *beyond* the individual's control, called effort (occupational choice, number of hours worked or investment in human capital) and circumstances (socioeconomic and cultural background or race), respectively. As a consequence, total inequality can be seen, in reality, as a combination of inequality of effort (IE) and inequality of opportunity (IO). Thus, an equal-opportunity policy must guarantee that those who exert an equal degree of effort, regardless of their circumstances, are able to achieve equal levels of outcome (i.e., the policy should *level the playing field*).

Recent contributions by the World Bank (2006), Bourguignon et al. (2007a) and Marrero and Rodríguez (2010) have noted that IO, in addition to being the one type of

¹ Dworkin (1981a and 1981b) took the issue a little further. This author argued that people should be held responsible for their preferences but not their resources. However, some philosophers (e.g., Arneson, 1989; Cohen, 1989) have criticized the separating line between those aspects for which a person should be held accountable (preferences) and those for which he should not (resources).

² See also Van de Gaer (1993).

inequality that is truly important from the standpoint of social justice, could exert a different effect (i.e., negative) on growth than IE, whose impact would be positive.³ Thus, correcting a country's IO would not only result in a fairer society in terms of social equality, but it would also spur economic efficiency and growth.

Given the importance of IO, the main goal of this paper is to measure and compare IO estimates among European countries using a homogenous database. In particular, we compute total income inequality and IO for 23 European countries in 2005 using the Survey on Income, Social Inclusion and Living Conditions in Europe (EU-SILC) database. Data requirements for comparing IO across countries in a homogenous way are severe (Lefranc et al., 2008). In this regard, the EU-SILC is an exception that gives information on individual disposable income and a rich set of circumstances (for its 2005 wave). This paper thus contributes to the existing literature by using a homogeneous database that combines a rich set of comparable circumstance variables for a large number of countries. IO is estimated as the between-type (*ex-ante*) inequality component following the parametric procedure of Ferreira and Gignoux (2011). This approach allows for the inclusion of the large set of circumstances in the EU-SILC even in the presence of small sample sizes and, consequently, avoids the lack of accuracy of alternative non-parametric estimates under these conditions.

In general, we find that Nordic (Denmark, Finland, Norway and Sweden), Western continental (Germany, Netherlands, Austria, Belgium and France) and some among the richer Eastern EU (Slovakia, Czech Republic, Slovenia and Hungary) countries are within the low-IO group. The high-IO group basically consists of Mediterranean (Italy, Greece and Spain), Atlantic (Portugal, Ireland and the UK) and poorer Eastern EU

 $^{^{3}}$ See Galor (2000, 2009) for an overview of the modern perspective on the relationship between inequality and economic development.

(Estonia, Latvia, Poland and Lithuania) countries. Moreover, although the IO and total inequality rankings are highly correlated, we note that some countries' ranks change significantly depending on whether IO or overall inequality is considered. For example, Sweden, Slovenia, Belgium, France, Ireland and Portugal rank worse in terms of IO than total inequality, while the opposite is true for Finland, Germany, Latvia and Slovakia.

In addition to these IO estimates, we would like to know which specific national characteristics have a causal effect on IO. But addressing these questions is quite challenging because, among other things, a sufficiently large and rich panel of data would be required. Unfortunately, our database (the EU-SILC) consists only of a cross-section of 26 countries for just one year (2005). Nevertheless, we conduct a more modest analysis and measure the degree of correlation between income inequality in 2005 (our IO and total inequality estimates) and a set of past and contemporaneous economic factors related to the degree of development, labor market performance, investment in human capital and social protection spending. Given the increasing importance of the topic, the current state of the art and the limited availability of data, we believe that a better understanding of cross-country differences in IO deserves this attempt.

The paper is structured as follows. Section 2 reviews the previous literature on IO. Section 3 presents the methodology employed, while Section 4 introduces the database used to measure IO in Europe. Section 5 shows the estimates found for IO in Europe. Section 6 shows results for correlations between IO and factors related with development, labor market, human capital investment and social policy. Finally, Section 7 offers some public policy recommendations based on our findings.

2. Revision of the literature

The modern theories of justice recognize that an individual's income is a function of variables *beyond* and *within* the individual's control, called circumstances and effort, respectively.⁴ As a consequence, overall inequality is the result of heterogeneous circumstances (IO), which represent individual initial conditions, and efforts (IE), which represent individual decisions.

There exist many procedures to estimate IO and this Section reviews the most relevant. A first distinction is made between the pioneering approaches of Roemer (1993) and Van de Gaer (1993). Roemer's procedure states that there is equality of opportunity if all individuals who exert the same degree of effort obtain the same level of outcome. For this task, he proposes to compute, across types, the minimum outcome level of individuals who exert the same degree of effort (i.e. individuals in the same *tranche*) and then maximize the average. Alternatively, Van de Gaer focuses on the set of outcomes available to individuals sharing similar circumstances (the opportunity set). Then, there is equality of opportunity if the opportunity set available to every individual does not depend on one's initial circumstances. As an equality-of-opportunity criterion, he proposes first averaging outcomes across types.⁵

These two alternative approaches have given rise to the so-called *ex-post* and *ex-ante* procedures, respectively (Fleurbaey, 2008). In sum, for the *ex-post* approach there is equality of opportunity if all individuals who exert the same effort obtain the same outcome, while for the *ex-ante* approach there is equality of opportunity if all

⁴ Talent could be considered a circumstance, however, this variable is controversial as it might reflect past effort of a person (while being a child) and hence is not obviously something for which a person should not be held accountable. Lefranc et al. (2009) also considers luck as an additional source of income.

⁵ Note that intergenerational mobility is a closed related concept to equality of opportunity if parental income is considered as the relevant circumstance (O'Neill et al., 2000; Van de Gaer et al., 2001).

individuals face the same set of opportunities regardless of their circumstances. We focus our attention in this paper only on the *ex-ante* approach.⁶

Roemer and Van de Gaer use the minimum function to comply with the Rawlsian maximin principle. However, other authors have followed alternative routes. On one hand, partial equality-of-opportunity orderings have been proposed. For example, Peragine (2002 and 2004) proposed the use of the Generalized Lorenz Curve to make ordinal welfare comparisons for income distributions according to equality of opportunity; Rodríguez (2008) proposed an equality-of-opportunity partial ordering based on the TIP's dominance criteria;⁷ Lefranc *et al.* (2009) considered a mechanism to contrast equality of opportunity based on the first and second stochastic dominance criterions in a model that considers circumstances, effort and also luck (see also Peragine and Serlenga, 2008). On the other hand, complete equality-of-opportunity orderings based on inequality indices have also been proposed. For example, Moreno-Ternero (2007) proposed to minimize the average of outcome inequality (across types) at each relative effort level;⁸ Lefranc et al. (2008) considered an index to measure inequality of opportunity based on the Gini index; Rodríguez (2008) provided a class of inequality-of-opportunity measures based on the Foster-Greer-Thorbecke family of poverty measures (Foster et al., 1984);9 Pistolesi (2009) used counterfactual distributions built on duration models to measure equality of opportunity.

In line with the last set of approaches, and given the importance of assessing the magnitude of IO in terms of overall inequality, the procedure of decomposing total

⁶ See Ooghe et al. (2007) and Fleurbaey and Peragine (2009) for a theoretical comparison between the ex post and ex-ante approaches.

⁷ The TIP curve is applied in the poverty literature, see for example Jenkins and Lambert (1997).

⁸ He also proposed to minimize the maximum inequality throughout the different levels of relative effort and the inequality between the average outcome of each type of individual.

⁹ It is worth noting that the first two mechanisms developed by Moreno-Ternero (2007) are particular cases of the class of measures proposed by Rodríguez (2008).

inequality into IO and IE components has gained great popularity in recent years. First proposed by Ruiz-Castillo (2003) and subsequently improved by Checchi and Peragine (2005 and 2010) and Ferreira and Gignoux (2011), overall inequality can be decomposed into two components, one due to IO and the other due to IE.¹⁰ Using an exante criterion, population is partitioned according to individuals' circumstances and IO is evaluated in terms of differences *between* individuals endowed with the same circumstances, so that IO is represented by the between-group component of overall inequality.¹¹

Among the alternative estimation procedures, a last distinction is made based on how IO and IE are finally estimated: non-parametrically (Checchi and Peragine, 2005 and 2010) or parametrically (Bourguignon et al., 2007b and Ferreira and Gignoux, 2011). In contrast to the standard non-parametric approach, the parametric method is a regression-based approach for computing the share of IO.¹² Nevertheless, the suitability of both estimation methods (parametric and non-parametric) depends mainly on the characteristics, the sample size and the observed circumstances of the database. When the number of observed circumstances is high and the sample size is not large enough, some group types may present a small number of observations and, as a result, the non-parametric estimates may be inaccurate. Meanwhile, the parametric approach assumes a particular specification, and the possible existence of relevant unobserved circumstances

¹⁰ A somewhat related decomposition is proposed in Cogneau and Mesplé-Somps (2009).

¹¹ Using an ex-post procedure, population is firstly partitioned into types, according to individuals' circumstances, and then each type is further subdivided according to personal effort. Correspondingly, IO is evaluated in terms of outcomes of individuals who have exerted the same effort, so that IO is represented by the *within-group* component of overall inequality.

¹² The main difference between the approaches in Ferreira and Gignoux (2011) and Bourguignon et al. (2007) is that the former seeks to estimate a lower-bound of the true IO because all individual circumstances certainly cannot be observed, while the latter seeks to estimate the effect of a specific (observed) set of circumstances by using Monte-Carlo simulations in order to estimate bounds around the possible biases in specific coefficients.

-correlated with the observed ones- may cause the residuals of the parametric regressions not to be orthogonal to the regressors.

In this paper we estimate the IO of 23 European countries in 2005 by using the EU-SILC database. Because this database contains a considerable number of circumstances, we apply the parametric (ex-ante) approach proposed in Ferreira and Gignoux (2011). In this manner, we can take advantage of all the circumstances included in the EU-SILC database and avoid the lack of accuracy in the non-parametric estimates. In the next section, we explain the method and in Section 4 we present the data and the set of circumstances that are used in the empirical analysis.

3. A Methodology for computing IO

Based on Ferreira and Gignoux (2011), this Section presents the method used for computing IO. Consider a finite population of discrete individuals indexed by $i \in \{1,...,N\}$, the individual income, y_i , is assumed to be a function of the amount of effort, e_i , and the set of circumstances, C_i , that the individual faces, such that $y_i = f(C_i, e_i)$. Effort is treated as a continuous variable, while, for each individual *i*, C_i is a vector of *J* elements, each element corresponding to a particular circumstance. While circumstances are exogenous because they cannot be affected by individual decisions, effort is assumed to be influenced, among other factors, by circumstances. Consequently, individual income can be rewritten as $y_i = f[C_i, e_i(C_i)]$.

Population then is divided into *M* mutually exclusive and exhaustive groups (or types), $\Gamma = \{H_1, ..., H_M\}$, where all the individuals in the same group *m* have the same circumstances: $H_1 \cup H_2 \cup ... \cup H_M = \{1, ..., N\}$, $H_r \cap H_s = \emptyset$, $\forall r$ and *s*, and $C_i = C_k$, $\forall i \text{ and } k \mid i \in H_m \text{ and } k \in H_m, \forall m.$ Effort distribution for individuals of type *m* is denoted by F^m , and $e^m(\pi)$ represents the level of effort exerted by an individual in the π^{th} quintile of that effort distribution, with $\pi \in [0, 1]$. Given type *m*, the income level attained by the individual in the π^{th} quintile is denoted by $v^m(\pi) = y^m(e^m(\pi))$. In this manner, the order of incomes and efforts within each type coincide since, for a particular type, the income will be determined exclusively by the effort.¹³ Thus, there is equality of opportunity when individual's income is independent of his social origins (Bourguignon et al., 2007b and Lefranc et al., 2008). Strictly speaking, this would translate into the following condition:

$$FF^{m}(y) = F^{k}(y), \forall m, k | H_{m} \in \Gamma, H_{k} \in \Gamma^{m}(y) = F^{k}(y), \forall m, k | H_{m} \in \Gamma, H_{k} \in \Gamma.$$
(1)

Given income distributions by types, first and second order stochastic dominance by types could be contrast. The stochastic dominance criterion, however, is partial and incomplete, since the distribution functions can cross (Atkinson, 1970). What is more, when the number of circumstances is large, the number of observations per type will be small, which, in practice, precludes an accuracy estimation of the distribution functions. An alternative is to consider a particular moment of said distributions, such as the average. Thus, given $\pi \in [0, 1]$, let us define

$$\mu = \left(\mu^{1}, ..., \mu^{M}\right) = \left(\int_{0}^{1} v^{1}(\pi) \, d\pi, ..., \int_{0}^{1} v^{M}(\pi) \, d\pi\right),\tag{2}$$

the *M*-dimensional vector of average incomes for the various types. A necessary (though not sufficient) condition to be equality of opportunity is that the elements of vector μ be equal, that is:

¹³ This property is equivalent to the *strictly increasing* axiom in the literature on IO (see O'Neill et al., 2000).

$$\mu^{m}(y) = \mu^{k}(y), \forall m, k | H_{m} \in \Gamma, H_{k} \in \Gamma.$$
(3)

As commented in the previous section, while Van de Gaer (1993) proposed maximizing the minimum average income, $Min(\mu) = \min_{m} \left\{ \int_{0}^{1} v^{m}(\pi) d\pi \right\}$, many other authors have proposed using an inequality index, such as the Gini or the Theil 0. In our context, one advantage of the latter proposal is that, by taking into account every element in the average vector μ and not just its minimum element, the calculation would be less subject to extreme values. Accordingly, IO can be defined as $I(\mu)$, where I is a specific inequality index.¹⁴

Of all the possible inequality indices that fulfill the basic principles found in the literature on inequality,¹⁵ Ferreira and Gignoux (2011) select the mean logarithmic deviation or Theil 0, T_0 , since it belongs to the Generalized Entropy class of inequality indices and, therefore, is additively decomposable (Bourguignon, 1979; Shorrocks, 1980; Cowell, 1980), has a path-independent decomposition (Foster and Shneyerov, 2000) and uses weights based on the groups' population shares. The decomposition of this index into between-group and within-group inequality components is

$$T_0(Y) = T_0(\mu) + \sum_{m=1}^{M} \frac{n^m}{N} T_0(y^m)$$
(4)

where n^m represents the population of type *m*. The between-group inequality index would be the IO index (actually, a *lower bound* of the IO –see below–), since the groups would be determined just by the observed individual circumstances. As for the within-

¹⁴ Note that whenever total inequality can be additively decomposed by population groups according to a set of circumstances, the IO term will be the *between-group* inequality component, while the *within-group* inequality component could be interpreted as the IE term. ¹⁵ The principle of progressive transfer

¹⁵ The principle of progressive transfers, symmetry, invariance to changes in scale and replication of the population (Cowell, 1995; Sen and Foster, 1997).

group inequality, this could be considered as that due to effort. However, this term may contain other elements arising from non-observed circumstances and/or luck. That is why our analysis focuses on aggregate inequality and on IO estimates.

As discussed in the previous Section, the between-group component can be nonparametrically estimated, but this approach presents serious problems of accuracy when the number of circumstances is high, as in our case. Therefore, parametric techniques should be used to yield reliable estimates. Following Bourguignon et al. (2007b) and Ferreira and Gignoux (2011), the parametric specifications rest on the assumption that the income of individual *i* is $y_i = f(C_i, e_i(C_i, u), v)$, where *u* and *v* represent random variables, like luck, as well as possible non-observed factors. If we now consider the reduced form of the above expression, $y = \Phi(C, \varepsilon)$, we can estimate the log-linear equation using ordinary least squares (OLS):

$$\ln y = C\lambda + \varepsilon . \tag{5}$$

Thus, once the within-group dispersion is accounted for, the OLS estimate would yield an approximation $\hat{\mu}_i = \exp[C_i \hat{\lambda}]$ for the individual incomes. Based on these estimated individual incomes, we can directly obtain the vector $\hat{\mu} = (\hat{\mu}^1, ..., \hat{\mu}^M)$, which is a

parametric version of the vector μ . Lastly, we compute IO as $IO = T_0(\hat{\mu})$.

4. The EU-SILC European database

The availability of suitable data is crucial to a rigorous study of IO. An empirical analysis of IO requires not only comparable measures of individual disposable income, but also for individual circumstances or social origins to be measured in a comparable

and homogeneous way. Unfortunately, there are few databases with this information, and even then, the number of circumstances tends to be limited.¹⁶

The database used in this paper is the EU Survey on Income, Social Inclusion and Living Conditions, or EU-SILC. This survey was recently implemented (in 2004), and only the data for 2005 is of use for our purposes, since this is the only year for which information is available on the parents' occupation and level of education, which are the most widely used variables to measure the individual social background in the related inequality-of-opportunity literature.¹⁷ Annual incomes always include transitory variations and measurement errors, and as a result, income averages for a given number of years could be useful in neutralizing these erratic components (Pistolesi, 2009). Unfortunately, we cannot average incomes because the EU-SILC is only available for our purposes in 2005, which is clearly the main disadvantage of this database.¹⁸

On the contrary, an initial benefit of this survey is that it offers information for a large number of countries (26 total), which gives its database sufficient heterogeneity in terms of economic features and public policies. The countries we use are: Austria (AT), Belgium (BE), Czech Republic (CZ), Germany (DE), Denmark (DK), Estonia (EE), Greece (EL), Spain (ES), Finland (FI), France (FR), Hungary (HU), Ireland (IE), Italy (IT), Lithuania (LT), Latvia (LV), the Netherlands (NL), Norway (NO), Poland (PL), Portugal (PT), Sweden (SE), Slovenia (SI), Slovakia (SK) and the United Kingdom

¹⁶ For example, the set of papers in Volume 13 of 2010 of the Review of Economic Dynamics consider databases with information on individual income; however, they do not contain information on individual circumstances. Likewise, studies on inequality of opportunity (Roemer et al., 2003, Rodríguez, 2008, Lefranc et al., 2009, Cogneau and Mesplé-Somps, 2009, and Ferreira and Gignoux, 2011) have based their results on the use of a different database for each country so a real cross-country analysis of inequality of opportunity has not been conducted yet.

¹⁷ See, for example, Roemer et al. (2003), Checchi and Peragine (2005 and 2010), Bourguignon et al. (2007b), Rodríguez (2008), Lefranc et al. (2009) and Ferreira and Gignoux (2011).

¹⁸ Nevertheless, it is worth noting that removing transitory income variations might lead to a smoothing of the role of effort, which might then overestimate the relative importance of IO. To neutralize for data extremes, we have omitted those observations with negative or zero incomes, and/or incomes 15 times higher than the mean income of their distribution.

(UK).¹⁹ A second advantage is the considerable number of circumstances that this database contains. For our study, we use the educational levels and occupations of both parents, the origin (national, European or rest of the world) of the individual and, lastly, a qualitative variable that measures the prevalent economic conditions in the individual's home during his/her childhood. To the best of our knowledge, the 2005 EU-SILC database features the highest number of individual circumstances measured homogeneously for such a large number of countries.

The variable used to calculate inequality is the disposable equivalent income for those households whose head is between 26 and 50 years of age.²⁰ This way, we consider the cohorts with the highest proportion of employed persons and avoid the composition effect (individuals with different ages are in different phases of the wage-earning time series) while approaching the concept of permanent income (Grawe, 2005). In terms of the IO calculation, it must be noted that the circumstance vector observed is, by definition, a subset of the vector of all possible circumstances. The estimated IO values, then, will be a lower-bound of the true IO, and will increase with the number of circumstances observed (Ferreira and Gignoux, 2011).²¹ That is why, when measuring

¹⁹ The EU-SILC database also contains information on Luxembourg, Iceland and Cyprus, though we opted not to use these countries due to the peculiarities they pose and to their small size. The specified codes correspond with the codes provided by Eurostat.

²⁰ The equivalence scale used in this paper is the same as that used in the EU-SILC database. Specifically, the equivalence scale is $e = 1 + 0.5(N_{14^+} - 1) + 0.3N_{13^-}$, where N_{14^+} is the number of household members 14 years of age or older and N_{13^-} is the number of household members 13 years of age or younger.

²¹ In principle, it is possible for unobserved circumstances to be negatively correlated with the set of observed circumstances. If this is the case, IO estimates would be overestimated (see Bourguignon et al., 2007b). However, for parametric estimates, every time we include a new circumstance, whichever correlation it has with the set of observed circumstances, the explained variance of income does not decrease, i.e., the coefficient of determination of the regression is at least as high as it was before the inclusion of the new circumstance. In this sense we can always assure that our parametric estimates are a lower bound. It is not in this problem is not unique to a study of IO, however, and is seen in practically every field of economics. For example, an analysis of salary discrimination must face the problem of heterogeneity that is not explained by the individual characteristics observed. Worse yet, econometric modeling normally introduces a random variable to somehow account for all non-observed variables.

IO, it is important that a database containing sufficient information on the individuals' circumstances be employed.

We present the summary statistics of the selected set of circumstances in Table 1. Given the restrictions imposed on the observations, it is remarkable that the sample size is larger than 2,500 units on average (the range goes from Latvia with 1,159 observations to Italy with 8,638 observations). First of all, we notice that the average of disposable equivalent household income ranges from about 3 thousand Euros for Lithuania, Poland and Slovakia, to the almost 28 thousand Euros of Norway, followed closely by the almost 27 thousand of the UK and the 25 thousand of Denmark and Ireland.²² It is worth noting that the difference between Denmark, the UK and Ireland is that the former shows a standard deviation which is more than half those of the UK and Ireland. These differences will be reflected in the inequality indexes estimated in the next section. In general, Nordic and Western continental countries present the highest disposable equivalent household income. Italy, Spain, Greece, Slovenia and Portugal follow the leading group, with an average personal income between 10 and 17 thousand. Finally, the remaining Eastern EU countries (Czech Republic, Estonia, Hungary, Lithuania, Latvia, Poland and Slovakia) are in the group of low-income countries, and their average income ranges from 3 to 5 thousand Euros.

By circumstances, we find that the greatest heterogeneity is observed for the levels of education attained by parents, especially in terms of their primary and secondary education. For example, the percentage of fathers with at least secondary education (most common in most countries) varies between 7% and 25% in Portugal, Spain, the UK, Ireland and Greece, up to the range of 70%-95% in Slovakia, Hungary, Finland,

²² These income variables are not PPP-adjusted, hence strict comparisons are misleading. Nevertheless, countries are in general ranked as expected.

Denmark, Czech Republic and Austria. For its part, the father's tertiary education exceeds 20% in Belgium, Denmark, Sweden, Estonia, Finland, Germany, Latvia, Lithuania, the Netherlands, Norway, Slovakia and the UK, while it is lower than 10% in Austria, Greece, Italy, Portugal, Poland and Slovenia. Similar profiles are found for the mother's education.

The distribution of the father's occupation (we do not have data for Sweden for this series) is more homogeneous across countries.²³ With the exceptions of Ireland, Estonia, Finland, Latvia, Greece and the Netherlands, the most common profession (with an average of 26% and a standard deviation of 4.6 points) is that of 'craft and related trades workers'. The next most common occupation on average (16%) is that of 'plant and machine operators and assemblers'. Among the various occupations considered, the one showing the greatest heterogeneity among countries is that of 'skilled agricultural, forestry and fishery workers', which varies between 20% and 35% in Finland, Portugal, Poland and Greece, and is less than 5% in Belgium, Czech Republic, Estonia, Ireland, Latvia, the Netherlands, Slovakia and the UK.

Regarding the economic perception during childhood (we do not have data for Austria, France, Germany, Greece and Portugal for this series), the most common response (on average) is 'never', with 37%; the 'rarely' and 'occasionally' answer reach just over 20%. However, there are also important differences between countries. For example, in Belgium, Denmark, the Netherlands, Norway and Sweden, almost 80% say they 'never' or 'rarely' had economic difficulties, while this percentage drops below 50% in Estonia, Italy, Lithuania, Poland, Slovakia and Slovenia. Finally, regarding the country of birth,

²³ We have considered the father's occupation as the relevant circumstance for most countries, given the large group of missing observations for the mother's occupation. The exception is the UK, where we have used the mother's occupation, because of the many missing observations for the father's occupation.

over 90% of individuals in the sample were born in their country of residence. Only Ireland has a significant percentage (nearly 10%) of people born in another EU country, while in the UK, Sweden, Latvia, Slovenia, Austria, Estonia and France, the percentage of residents born outside the EU is between 8% and 13%.

INSERT TABLE 1 ABOUT HERE

5. Inequality of Opportunity in Europe

In this section we provide overall inequality and IO estimates based on the ex-ante parametric approach of Ferreira and Gignoux (2011) described in Section 3. As a first step, we estimate (by OLS) the regression $\ln y = C\lambda + \varepsilon$ for each country, which relates household income (in logarithms) with the set of circumstances considered in the analysis. The reduced-form OLS regression estimates for all 23 European countries are presented in Table 2.²⁴ In general terms, coefficients have the expected sign.

The parents' education has a positive influence on children's income, which increases with the educational level of the father and/or the mother. In general, with respect to the omitted category (parents with less than primary education), results are specially significant and robust when parents attain at least secondary or tertiary education. Both variables, father's education and mother's education, are highly significant in France, Germany, Greece, Italy, Portugal, Spain and the UK. However, for some countries (Belgium, Estonia, Hungary, Ireland, Poland and Slovakia), the education attained by

²⁴ When an explanatory variable's estimated coefficient is not shown, that is because there are no observations with that circumstance in the sample. As emphasized by Ferreira and Gignoux (2011), since this is a reduced-form equation, estimates cannot be interpreted causally, and coefficients would capture not only the direct effects of circumstances on income, but also the indirect effects on income through non-included circumstances or effort.

the mother seems to be more significant than the education attained by the father, while the opposite is true in Latvia, Lithuania, the Netherlands, Sweden and Slovenia.

Regarding the occupation of the father, and taking 'workers in the farming, forestry and fishing' sectors as a reference, all of the remaining occupations tend to have a positive effect on the individual's income. The exception is that of the 'Elementary occupation' concept, whose relative effect is sometimes negative, although just significant for Austria, Hungary and Italy. Among the alternative occupations, the most robust results are found for the 'Managers' category, followed by that of 'Technicians and Professionals', although some exceptions can be found in Belgium, Denmark, the Netherlands and Norway.

The perception of having 'financial difficulties during the childhood years' tends to have a negative effect on household income. Since the omitted category is that the individual 'never had difficulties', most of the estimated coefficients for all other categories are negative, though the number of significant coefficients associated with these categories is smaller than those found for the parents' education variables. Finally, a circumstance that also tends to have a negative effect on household income is that of having roots outside the country of residence, especially if the country of origin is not European. Given the reference category "be born in the country of residence", being from another EU country is insignificant in most cases (except for Belgium, Italy, Portugal and Spain, where it is negative, and Slovakia, where it is positive), while being born outside the EU constitutes a significant and negative circumstance in explaining household income.

INSERT TABLE 2 ABOUT HERE

17

Now, Table 3 shows the main results for income inequality (Theil 0) and IO for the 23 European countries considered. The first row contains the estimates of overall inequality, the second row the IO estimates, the third provides the relative IO measure, i.e., the IO to total inequality ratio²⁵, the fourth and the fifth rows show the position of each country (from lowest to highest) by Theil 0 and IO, respectively, and the last row provides the number of observations used to calculate these indexes. Moreover, we show below each estimate the corresponding standard error estimated by the bootstrapping method using the formula (Davison and Hinkley, 2005):

$$\hat{\sigma}(\hat{T}) = \sqrt{\frac{1}{R-1} \sum_{r=1}^{R} \left(T^* - \bar{T}^* \right)^2}, \qquad (6)$$

where T is the corresponding index and R is the number of replicates.²⁶

INSERT TABLE 3 ABOUT HERE

Since the database is homogenous, the set of circumstances (for most countries) and the sample design are common to all countries, and our inequality measures can be used to compare cross-country differences in terms of (absolute and relative) IO. It is worth noting that, despite the required characteristics of our selected sample (recall from the previous section), we observe that the ranking of our overall inequality (Theil 0) estimates is quite similar to that induced from the Eurostat estimates. In fact, their linear coefficient of correlation is 0.92. According to the Eurostat Gini index in 2005, the

²⁵ This index depends, by construction, not only on opportunities but also on the IE component. For instance, if total inequality increases due to a higher IE component, the relative IO index would decrease, though absolute IO has not changed. As a result, the use of this relative index is problematic. Despite this shortcoming, and in order to check the robustness of our results, we have also considered the relative IO index.

²⁶ For our calculation, we assumed R = 1000. Cowell and Flachaire (2007) find that, in general, the bootstrap technique improves the numerical performance of the significance tests. Moreover, for small sample sizes, this technique yields a closer margin to the nominal confidence intervals (Davison and Hinkley, 2005).

lowest inequality is observed in Sweden, Denmark and Slovenia, with Gini levels of 0.23-0.24, closely followed by Czech Republic, Finland, Germany, Austria, Slovakia, Netherlands, Hungary, France, Belgium and Norway, with Gini estimates between 0.26 and 0.28.²⁷ All other European countries present clearly higher Gini indexes (at least 15% higher), between the 0.32 of Ireland and Spain and the highest levels of 0.36-0.38 in Poland, Lithuania, Latvia and Portugal.

Figure 1 shows our Theil 0 estimates, together with the estimated bootstrapped standard deviations (using one standard deviation around the point estimate). Countries are sorted from lowest to highest Theil 0 estimates. We can see clearly that the two main groups (low- and high-inequality countries) are equivalent to those provided by Eurostat, though there are some minor differences when looking inside each group. Nevertheless, considering the fact that some standard deviations overlap, these within-group differences are in some cases not relevant.²⁸

INSERT FIGURE 1 ABOUT HERE

The rest of the Section is devoted to commenting on the results of IO. As noted in Section 2, we could apply a partial ordering (see Peragine, 2002 and 2004, Rodríguez, 2008 and Lefranc et al., 2009) to measure IO. The advantage of an ordinal criterion is that comparisons of IO between countries would be more robust. However, an ordinal criterion will be not conclusive in many cases. For example, if there is no first and second stochastic dominance when applying the method proposed in Peragine (2002 and 2004) and Lefranc et al. (2009), or if inequality-of-opportunity curves cross when applying the method proposed in Rodríguez (2008), we would be unable to conclude

²⁷ Data from Eurostat 2005, in the 'Living conditions and welfare statistics' section (Gini coefficients): http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=ilc_di12&lang=en.

²⁸ For example, for the low-inequality group, the estimates for Belgium, Sweden, France, Finland, Norway, Austria and Czech Republic clearly overlap; for the high-inequality group, there is overlap, on the one hand, between Italy, Estonia and UK, and, on the other hand, between Greece, Spain and Portugal.

which country presents a higher IO. For this reason, we opted to compute a complete ordering based on the mean logarithmic deviation, thus enabling us to compare the IO for all countries. Nevertheless, we have considered in our comparisons the fact that some standard deviations overlap.²⁹

Figure 2 shows the IO estimates together with their standard deviations. Countries are now ranked from smallest to largest IO. As was the case when comparing Theil 0 estimates, IO intervals overlap for some countries. As a result, we find again two main groups of countries, which basically coincide with those groups of overall inequality: low-IO countries (Denmark, Germany, Sweden, Finland, the Netherlands, Belgium, Slovenia, France, Czech Republic, Austria, Slovakia, Hungary and Norway) and high-IO countries (Latvia, Poland, Lithuania, Portugal, Spain, Greece, the UK, Estonia, Italy and Ireland). The first group basically comprises Nordic, Continental and some Easter countries. In contrast, the second group basically consists of Mediterranean, Atlantic and some other Eastern countries.

We find numerous similarities when comparing these results with previous studies. Based on a heterogeneous database of 11 countries constructed by Roemer et al. (2003), Rodríguez (2008) and Lefranc et al. (2009) applied their proposals.³⁰ In general terms, these authors find that Denmark (1993), Sweden (1991), Norway (1995) and Germany (1994) are the countries with lowest IO, and Italy (1993) and Spain (1991) are the ones with highest IO. The remaining European countries (the Netherlands, Belgium, France and the UK) present an intermediate IO. Therefore, considering the results of previous

²⁹ We are aware that a robust ranking comparison between countries would require the application of a dominance criterion, which goes beyond the scope of this paper.

³⁰ Those countries were originally the following: Belgium (1992), Denmark (1993), France (1994), Great Britain (1991), Italy (1993), The Netherlands (1995), Norway (1995), Spain (1991), Sweden (1991), the United States (1991) and West Germany (1994). Rodríguez (2008) also included Spain (2005), while Lefranc et al. (2009) did not consider the data for Denmark (1993) and Spain (1991).

studies together with our results, we can say that the IO ranking of European countries has changed little in the last fifteen years.

Although IO intervals overlap in some cases, we note that some countries' ranks change significantly depending on whether overall inequality or IO is considered. For example, Sweden, Slovenia, Belgium, France, Hungary, Ireland and Portugal rank worse in terms of IO than total inequality, while the opposite is true for Finland, Germany, Latvia and Slovakia.

INSERT FIGURE 2 ABOUT HERE

Lastly, we comment on the results for the relative IO, i.e., the IO to total inequality ratio (see Table 3). For illustrative purposes, we show the ranking of countries from lowest to highest relative IO in Figure 3. We first notice that the percentage of total inequality represented by IO in Europe is on average approximately 9%, ranging from 2% in Denmark to 22% in Portugal.³¹ Comparing absolute with IO ratio estimates, most countries maintain their relative position. However, there are several notable exceptions. When looking at IO ratios instead of absolute estimates, the positions of Slovenia, Hungary, Ireland and Belgium worsen significantly, while Latvia, Greece and Poland perform better.

INSERT FIGURE 3 ABOUT HERE

³¹ In this respect, note that Checchi and Peragine (2005 and 2010) computed an IO ratio below 10% for Italy, while Ferreira and Gignoux (2011) found percentages between 20% and 33% for six Latin American countries (Brazil, Colombia, Ecuador, Guatemala, Panama and Peru) when using the income variable.

6. Accounting for Inequality of Opportunity in Europe

The *augmented Kuznets curve hypothesis* (Milanovic, 1994; Tanzi, 1998) emphasizes that inequality is determined by '*given*' (*long-term*) factors involving the country's resources, the degree of development, social norms, etc., and by *short-term* factors, such as education, the functioning of the labor market, social public expenditures, etc.³² Although we are aware that our limited sample prevents us from carrying out a rigorous quantitative analysis of the determinants of IO, we still consider it instructive to show the correlation between different measures of inequality (total and IO) and certain indicators related to those long- and short-term factors.³³

For this exercise, we consider a variety of indicators that reflect a country's level of development: the PPP-adjusted per capita GDP and the percentage of jobs concentrated in the agriculture and in the service sector. With respect to the labor market, we take the employment rate (total and female) and the unemployment rate (total and the long-term rate). We also consider standard variables for measuring education: the population with at least a secondary level of education (total and female) and with at least a university degree (both as a percentage of the population older than 15), and the percentage of the population between the ages of 18-24 without a secondary education degree (early leaves). Finally, we consider the total spending on social protection, as well as their

³² Taking a set of European regions and using a simple cross-section regression equation, Perugini and Martino (2008) study the factors that explain aggregate inequality in Europe, distinguishing between these long- and short-term factors.

³³ Total inequality and IO are highly persistent variables. As a result, it is possible that past levels of either one of these measures of inequality cause past or present levels of education, economic development or social public expenditures. In addition, omitted (observable or not) variables, such as the quality of institutions or the initial economic efficiency, might affect not only the employment rate, social protection and education attained, but also current inequality. It is thus needless to say that reverse causation and omitted bias problems (both sources of endogeneity) are difficult to address even in a dynamic panel data model when considering a cross-section regression with just 23 observations. We acknowledge this point to an anonymous referee.

various outlays including child care, disability, social exclusion, health care, pensions and unemployment (all measured as a percentage of GDP).³⁴

Table 4 shows contemporaneous (i.e., inequality and indicators are both measured in 2005) and lagged (i.e. inequality is measured in 2005 and indicators are measured in 1998) cross-correlations of total inequality and IO with respect to all the indicators commented above. It seems interesting to calculate cross-lagged correlations and compare them with contemporaneous correlations. The reason lies in the fact that most of the short- and long-term factors may have effective effects on inequality and IO with some delay. For example, education, health care or long-term unemployment policies are clear examples of the existence of such delays on their possible effects on income and hence on income inequality. Although the differences are generally small because of the persistence of inequality and IO, several cases are worth noting. For example, the lagged correlation is clearly superior in magnitude to the contemporary one for the cases of overall and long-term unemployment rate, education dropouts, expenditures on social protection, unemployment and health care expenditures. We focus on lagged correlations hereinafter.

INSERT TABLE 4 ABOUT HERE

The correlations between both measures of inequality and development indicators are negative.³⁵ This negative correlation is explicitly shown in Figure 4, which represents the scatter plot between IO and per capita real GDP (PPP). This result is consistent with the down slope of a Kuznets inverted U curve. Unfortunately, an exhaustive analysis of the Kuznets hypothesis requires more data points, which goes beyond the possibilities of the EU-SILC database. Nevertheless, we can use Figure 4 to show differences across

³⁴ The sources used for these indicators and their descriptive statistics are shown in the Appendix.

³⁵ Note that the correlation with the share of employment in the agriculture sector is positive, but this indicator must be taken as one of non-development.

countries in IO levels for a given degree of development. First, eastern European countries with higher per capita GDP (Hungary, Slovakia, Czech Republic and Slovenia) are well below the regression curve, meaning that their IO levels are below what would be associated with their levels of development. However, poorer eastern EU countries (Estonia, Latvia, Poland and especially Lithuania) do not perform so well. Thus, while their Communist roots and the great opportunities created in these economies after the expansion of the European Union could partly explain a favorable situation for IO, other factors inherent to each economy (education, social policies, institutional changes, etc.) must also be playing an important role. Focusing on the richest western countries, there are three clearly distinguishable groups: Denmark, Finland, Austria, Germany, Norway and the Netherlands, whose IO levels are less than expected based on their levels of development; the UK, Italy, Ireland, Spain and especially Portugal, whose IO levels are clearly higher than expected; and, Sweden, France, Belgium and Greece, which are very close to the regression line. Despite having highlighted certain geographical patterns among European countries, much remains to be explained in terms of the differences noted in total inequality and IO for these countries, which is a challenging and promising extension of the paper.

From a theoretical standpoint, the correlation between the labor market and inequality is complex and inconclusive (Burniaux et al., 2006). On the one hand, better functioning of the labor market involves less exclusion, and therefore less inequality. This result could be applied to IO if the labor market favored the inclusion of those population sectors that had, a priori, fewer opportunities, such as immigrants, youth and women. On the other hand, labor inclusion could place pressure on less-qualified employees as a whole, increasing salary differences between this group and that of more qualified workers (Topel, 1994). In general, we find negative correlations between inequality (total and IO) and labor market indicators, although they are the weaker among all analyzed.³⁶ Results for the unemployment rate and long-term unemployment rate deserve some attention. On one hand, their lag correlations with respect to total inequality are close to 0.5. On the other hand, while IO is poorly lag correlated with the overall unemployment rate (0.3), its lag correlation with long-term unemployment is close to 0.5. Accordingly, it seems that while unemployment may affect overall inequality, long-term unemployment, which is basically supported in a higher proportion by people with poor opportunities, is the one that matters for IO. For illustrative purposes, Figure 5 shows the scatter plot between IO and the long-term unemployment rate.

With respect to the education variables, the theory says that higher levels of education help to balance the initial distribution of human capital, and thereby reduce one of the main causes of inequality in developed economies (Tanzi, 1998).³⁷ With the exception of the variable for tertiary and upper education attained, our estimated correlations are consistent with this idea.³⁸ Moreover, correlations are higher for IO than for total inequality. Among all estimates, the correlation between IO and past dropouts is the closest to one. Accordingly, it seems that preventing dropouts and promoting the attainment of a secondary level of education should be particularly helpful in balancing the distribution of human capital opportunity. Figure 6 shows the scatter plot between early school-leavers and IO. The fit is clearly positive, and it is the most significant among all the indicators considered. When compared to other scatter plots, we note the

³⁶ We have also calculated correlations between IO and part-time employment, temporary employment, self-employment and unemployment rate gaps but they are close to zero. For simplicity, these results are not shown in the table.

³⁷ Empirical evidence suggests that parents with higher levels of education generally attach a higher importance to the education of their children (Kirchsteiger and Sebald, 2010).

³⁸ Bergh and Fink (2009) find that the effect of elite education on income inequality is ambiguous because it decreases income dispersion at the bottom, and increases income dispersion at the top of the income distribution.

good fit for Greece, Spain, the UK, Slovenia, Latvia and Hungary, as well as the improvement for Ireland and the nearly perfect fit for Portugal. Although this result is by no means definitive proof of the relevance of this variable to explaining observed differences in IO across European countries, its significant correlation is worth noting.

Finally, with respect to social protection expenditures, all items are negatively correlated with total inequality and IO. It is worth noting that, for all social expenditure concepts, we observe very little differences between IO and Theil 0 correlations. Some of these social expenditures, together with those for education, are the most correlated factors with IO. Figure 7 shows the scatter plot between social protection spending (as a percentage of GDP) and IO. The fit is clearly negative and significant. An additional finding is that some items among those included in social protection expenditures are more correlated with inequality and IO than others. In particular, social exclusion, health care and, especially, child care expenditures are highly correlated with both inequalities, overall and IO. In this respect, note that child care expenditure is a clear way to equalize opportunities between women and men, in particular because it improves the situation of those women with worse initial circumstances. The remaining items (unemployment benefits, pensions, disability and work leaves) are less correlated with both concepts of inequality.

INSERT FIGURES 4, 5, 6 AND 7 ABOUT HERE

7. Concluding remarks

The modern theories of justice recognize an individual's income as being a function of the effort made and of the initial circumstances affecting the individual. And yet, individuals are only responsible for their own efforts, since the circumstances remain beyond their control. Thus, a greater inequality in the distribution of income does not imply, *per se*, that the course of the economy in general, or the redistributive capacity of a public policy in particular, is bad. It may happen that the level of effort made by individuals is different. In fact, a country's fiscal policy could correct the uneven distribution of initial circumstances while at the same time respecting the individual labor supply. For this to happen, a public policy must be implemented that, far from simply redistributing income, provides every individual with the same initial conditions without modifying the economic incentives to maximize effort.

When comparing the IO results obtained in this paper for 23 European countries with previous IO results (obtained from heterogeneous databases), we observe a high persistence of the IO levels and rankings between countries. Thus, we find that the Nordic countries, continental countries and some eastern countries are low-IO countries, while the Mediterranean countries, Atlantic countries and some other eastern countries are high-IO countries. Understanding the main factors (institutional, political, cultural or geography) behind the persistence of such country differences constitutes, therefore, one of the main challenges in the agenda of the inequality-of-opportunity analysis.

As preliminary evidence, our findings highlight educational policies first and foremost. In particular, a reduction in the academic dropout rate seems to constitute a fundamental tool to increasing the opportunities available in an economy. Reaching secondary education levels would also help to reduce IO indices, while tertiary education does not seem to have a significant impact on IO. A second pillar on which any policy aimed at reducing IO seems to be based is social protection spending, though not all items of expenditure are equally correlated. Spending to reduce social exclusion and on child and health care are greatly correlated with IO, while expenses on unemployment benefits, retirement and disability show little correlation. Correspondingly, in order to reduce IO without affecting the public sector budget, properly tuning the composition of social expenditure could be a better policy than uniformly increasing public expenditure.

Apart from accumulating empirical evidence, it is clear from the above that future research on equality of opportunity should also focus its attention on developing a formal theoretical framework which explicitly specifies the IO factors and its channels of causality.

REFERENCES

— Arneson, R. (1989): "Equality and equal opportunity for welfare", *Philosophical Studies*, 56, pp. 77-93.

— Atkinson, A. B. (1970), "On the measurement of inequality", *Journal of Economic Theory*, 2, 244-263.

— Bergh, A. and Fink, G. (2009), "Higher education, elite institutions and inequality", *European Economic Review*, 53, 376-384.

— Bourguignon, F. (1979), "Decomposable income inequality measures", *Econometrica*, 47, 901-920.

— Bourguignon, F., Ferreira, F. and Walton, M. (2007a), "Equity, efficiency and inequality traps: a research agenda", *Journal of Economic Inequality*, 5, 235-256.

— Bourguignon, F., Ferreira, F. and Menéndez, M. (2007b), "Inequality of opportunity in Brazil", *Review of Income and Wealth*, 53, 585-618.

— Burniaux, J., F. Padrini and Brandt, N. (2006), "Labour market performance, income inequality and poverty in OECD countries", *OECD Economic Department Working Papers*, 500.

— Checchi, D. and Peragine, V. (2005), "Regional Disparities and inequality of opportunity: the case of Italy", *IZA Discussion Paper* 1874/2005.

— Checchi, D. and Peragine, V. (2010), "Inequality of opportunity in Italy", *Journal of Economic Inequality*, 8, 429-450.

— Cogneau, D. and Mesplé-Somps, S. (2009), "Inequality of Opportunity for Income in Five Countries of Africa", *Research on Economic Inequality*, 16, 99-128.

— Cohen, G. A. (1989): "On the currency of egalitarian justice", *Ethics*, 99, pp. 906-944.

— Cowell, F. A. (1980), "On the structure of additive inequality measures", *Review* of *Economic Studies*, 47, 521-531.

— Cowell, F. A. (1995), *Measuring Inequality* (2nd Edition). Hemel Hempstead: Prentice Hall/Harvester Wheatsheaf.

— Cowell, F. A. and Flachaire, E. (2007), "Income Distribution and Inequality Measurement: The Problem of Extreme Values", *Journal of Econometrics*, 141, 1044-1072.

29

— Davison, A. C. and Hinkley, D. V. (2005), *Bootstrap methods and their application*. Cambridge University Press, New York.

— Dworkin, R. (1981a): "What is equality? Part 1: Equality of welfare", *Philosophy & Public Affairs*, 10, pp. 185-246.

— Dworkin, R. (1981a): "What is equality? Part 2: Equality of welfare", *Philosophy & Public Affairs*, 10, pp. 283-345.

— Ferreira, F. and Gignoux, J. (2011), "The measurement of inequality of opportunity: theory and an application to Latin America", *Review of Income and Wealth* (forthcoming).

— Fleurbaey, M. (2008), *Fairness, responsibility, and welfare*. Oxford: Oxford University Press.

— Fleurbaey M. and Peragine V. (2009), "Ex ante versus ex post equality of opportunity", *Working Papers ECINEQ*, 141.

— Foster, J. E. and Shneyerov, A. (2000), "Path independent inequality measures", *Journal of Economic Theory*, 91, 199-222.

— Foster, J. E., Greer J. and Thorbecke, E. (1984), "A class of decomposable poverty measures", *Econometrica*, 52, pp. 761-65.

— Galor, O. (2000), "Income distribution and the process of development", *European Economic Review*, 44 (4-6), 706-712.

— Galor, O. (2009), "Inequality and economic development: an overview", Working Papers 2009-3, Brown University, Department of Economics.

— Grawe, N. (2005), "Lifecycle bias in estimates of intergenerational earnings persistence", *Labour Economics*, 13, 551-570.

— Jenkins, S. P. and Lambert P. J. (1997), "Three 'I's of poverty curves, with an analysis of UK poverty trends", *Oxford Economic Papers*, 49, pp. 317-327.

— Kirchsteiger, G. and Sebald, A. (2010), "Investments into education—Doing as the parents did", *European Economic Review*, 54, 501-516.

— Kuznets, S. (1955), "Economic growth and income inequality", *American Economic Review*, 45, 1-28.

— Lefranc, A., Pistolesi, N. and Trannoy A. (2008), "Inequality of opportunities vs. inequality of outcomes: Are Western societies all alike?", *Review of Income and Wealth*, 54, 513-546.

— Lefranc, A., Pistolesi, N. and Trannoy A. (2009), "Equality of opportunity and luck: Definitions and testable conditions, with an application to income in France", *Journal of Public Economics*, 93, 1189-1207.

— Lucas, J. R. (1995), *Responsibility*. Oxford: Clarendon Press.

— Marrero, A. G. and Rodríguez, J. G. (2010), "Inequality of opportunity and growth", *Working Papers ECINEQ*, 154.

— Milanovic, B. (1994), "Determinants of cross-country income inequality : an augmented Kuznets hypothesis", *Policy Research Working Paper Series*, 1246, The World Bank.

— Moreno-Ternero, J. D. (2007), "On the design of equal-opportunity policies", *Investigaciones Económicas*, 31, 351-374.

— O'Neill, D., Sweetman, O. and Van de Gaer, D. (2000), "Equality of opportunity and kernel density estimation: an application to intergenerational mobility", in *Advances in Econometrics* 14: Applying Kernel and Nonparametric Estimation to Economic Topics, T. B. Fomby y R. Carter (eds.), JAI Press: Stamford.

— Ooghe, E., Schokkaert E. and Van de Gaer, D. (2007), "Equality of opportunity versus equality of opportunity sets", *Social Choice and Welfare*, 28, 209-230.

— Peragine, V. (2002), "Opportunity egalitarianism and income inequality", *Mathematical Social Sciences*, 44, 45-60.

— Peragine V. (2004), "Ranking of income distributions according to equality of opportunity", *Journal of Income Inequality*, 2, 11-30.

— Peragine V. and Serlenga L. (2008), "Higher education and equality of opportunity in Italy" in J. Bishop and B. Zheng (eds.), *Research in Economic Inequality*, 16, 1-31.

— Perugini, C. and Martino, G. (2008), "Income inequality within European regions: determinants and effects on growth", *Review of Income and Wealth*, 54, 373-406.

— Pistolesi, N. (2009), "Inequality of opportunity in the land of opportunities, 1968-2001", *Journal of Economic Inequality*, 7: 411-433.

Rawls, J. (1971), A Theory of Justice. Harvard University Press, Cambridge, M.A.

— Roemer, J. E. (1993), "A pragmatic approach to responsibility for the egalitarian planner", *Philosophy & Public Affairs*, 10, pp. 146-166.

Roemer, J.E. (1996), *Theories of Distributive Justice*. Harvard University Press,
 Cambridge, M.A.

— Roemer, J.E. (1998), *Equality of Opportunity*. Harvard University Press, Cambridge, M.A.

— Roemer, J.E., Aaberge, R., Colombino, U., Fritzell, J., Jenkins, S., Marx, I., Page, M., Pommer, E., Ruiz-Castillo, J., San Segundo, M. J., Tranaes, T., Wagner, G. and Zubiri, I. (2003), "To what extent do fiscal regimes equalize opportunities for income acquisition among citizens?", *Journal of Public Economics*, 87, 539-565.

— Rodríguez, J. G. (2008), "Partial equality-of-opportunity orderings", *Social Choice and Welfare*, 31, 435-456.

— Ruiz-Castillo, J. (2003), "The measurement of the inequality of opportunities", *Research on Economic Inequality*, 9, 1-34.

Sen, A. (1980), "Equality of what?" in *Tanner Lectures on Human Values*, vol.I. Cambridge University Press.

— Sen, A. (1985), *Commodities and Capabilities*. Amsterdam. North Holland.

— Sen, A. and Foster, J. E. (1997), *On Economic Inequality*, Oxford University Press, Oxford.

— Shorrocks, A. F. (1980), "The class of additively decomposable inequality measures", *Econometrica*, 48, 613-625.

— Tanzi, V. (1998), "Fundamental determinants of inequality and the role of government". *IMF Working Paper*, 178.

— Topel, R. H. (1994), "Regional labor markets and the determinants of wage inequality", *American Economic Review*, 84, 17-22.

— Van de Gaer, D. (1993), "Equality of opportunity and investment in human capital", *Catholic University of Leuven*, Faculty of Economics, no. 92.

— Van de Gaer D., E. Schokkaert and Martinez, M. (2001), "Three meanings of intergenerational mobility", *Economica*, 68, 519-538.

World Bank (2006), World development report 2006: equity and development,
 Washington, DC: The World Bank and Oxford University Press.

Table 1. Summary statistics of the EU-SILC database.

	AT	BE	CZ	DE	DK	EE	EL	ES	FI	FR	HU	IE	IT	LT	LV	NL	NO	PL	РТ	SE	SI	SK	UK	Mean	Std
Selected sample size ⁽¹⁾	2,155	1,838	1,589	4,255	1,241	1,377	2,126	5,389	1,980	3,725	2,590	1,449	8,638	1,702	1,159	1,695	1,423	6,055	1,654	1,342	1,393	2,292	1,874	2,563	1,869
Equivalized personal incor	ne ⁽²⁾																								
Average	19,633	19,553	5,075	20,163	24,716	3,753	11,766	13,041	20,930	18,533	3,950	24,359	17,281	2,736	3,058	19,807	28,470	3,187	9,693	18,908	10,045	3,212	26,850	14,292	8,629
Standard deviation	10,283	9,848	3,089	12,609	8,628	2,515	8,230	8,187	12,135	9,858	2,595	15,978	10,711	1,970	2,856	8,762	13,742	2,527	7,954	7,474	4,246	1,934	18,361	8,021	4,720
Father's Education																									
Less than Primary*		12.6				0.5	23.7	21.3	0.2	5.3	0.6	1.9	12.3	4.8	1.1			10.6	30.2	1.6	3.2		44.6	10.9	12.9
Primary	0.1	27.0	0.4	2.1		11.3	51.6	56.1	1.9	50.9	13.8	62.1	50.8	32.8	11.0	21.7		37.6	58.9	41.3	36.5	4.7		28.6	21.9
Secondary	95.3	38.8	88.0	60.3	78.4	63.7	15.8	12.5	75.3	32.9	71.9	22.4	33.2	37.7	67.6	54.9	56.0	45.0	7.0	35.8	51.3	84.3	19.5	49.9	25.6
Tertiary	4.6	21.6	11.6	37.5	21.6	24.5	8.9	10.0	22.7	10.9	13.6	13.6	3.6	24.7	20.4	23.4	44.0	6.9	3.9	21.2	9.0	11.1	35.8	17.6	11.0
Mother's Education																									
Less than Primary*		14.2				0.9	28.6	24.3	0.2	6.3	0.9	1.7	16.4	5.7	1.5			12.0	41.0	2.6	4.4		53.9	13.4	16.0
Primary	1.8 ⁽³⁾	31.8	0.4	3.3	0.2	10.6	52.4	60.0	2.1	57.2	16.4	57.9	55.5	34.1	10.9	25.0		41.9	51.8	41.1	54.6	5.1		30.6	22.7
Secondary	62.0 ⁽³⁾	39.0	93.3	81.3	83.3	61.9	12.9	11.3	80.8	28.1	72.4	29.3	26.6	32.8	67.6	65.5	67.8	39.9	3.6	37.0	35.6	89.5	26.6	49.4	27.5
Tertiary	3.9 ⁽³⁾	15.0	6.2	15.4	16.5	26.6	6.1	4.3	16.9	8.4	10.3	11.2	1.5	27.4	20.0	9.5	32.2	6.2	3.6	19.3	5.5	5.4	19.5	13.1	8.5
Father's Occupation																									
Manager	5.0	11.5	4.3	6.7	9.6	10.5	11.7	6.7	10.2	8.5	5.9	25.1	9.9	6.6	6.2	23.1	13.2	3.5	7.3		4.7	8.3	13.4	9.6	5.5
Professional	3.4	12.9	6.7	16.6	13.3	8.3	4.5	3.9	8.0	10.7	6.7	9.8	3.7	8.1	9.2	14.2	10.5	4.8	2.5		5.2	7.7	15.4	8.5	4.1
Technician	12.4	6.8	16.1	12.2	10.8	5.4	2.2	4.8	12.9	8.4	5.4	3.0	7.5	2.9	5.8	15.2	18.9	6.5	4.3		9.9	11.0	8.2	8.7	4.6
Clerk	6.1	10.3	3.0	7.4	4.4	0.9	5.4	5.8	1.8	5.2	3.2	6.4	5.7	2.2	1.3	7.3	4.6	3.1	5.1		4.8	2.9	3.7	4.6	2.2
Salesman	10.5	6.1	3.9	3.0	6.2	1.2	4.8	7.8	3.8	3.3	3.3	5.6	4.2	2.0	2.2	4.5	4.6	2.2	5.7		5.7	2.8	3.8	4.4	2.1
Skill agricultural*	14.6	4.1	4.1	5.3	12.6	3.0	34.8	14.2	22.4	12.0	10.5	1.1	11.9	5.1	2.8	1.5	9.6	24.4	22.7		13.4	2.7	3.0	10.7	8.9
Craft trade	25.7	25.4	35.6	31.0	23.0	28.7	18.0	24.8	21.5	24.1	35.5	19.8	28.1	27.3	27.5	19.5	22.7	29.4	28.1		26.2	28.8	23.9	26.1	4.6
Machine operator	7.7	9.5	19.5	11.3	8.3	33.0	7.7	11.7	15.5	17.4	17.6	10.1	14.7	24.3	31.1	9.4	14.0	16.4	12.0		23.3	22.6	16.2	16.1	7.2
Elementary occup.	14.5	11.0	5.7	5.3	11.0	7.6	9.9	18.7	3.3	7.1	10.0	17.1	12.4	20.6	11.8	4.0	1.0	8.1	10.9		6.0	13.1	12.4	10.1	5.0
Armed/military	0.1	2.3	1.1	1.1	0.8	1.5	1.0	1.7	0.7	3.4	1.9	1.9	1.9	0.8	2.1	1.4	0.9	1.6	1.3		0.9	0.0	0.0	1.3	0.8
Economic Difficulties duri	ng childh	bod																							
Most of time		3.1	3.6		1.9	2.4		7.9	4.2		9.3	6.9	12.7	8.0	5.3	2.3	1.8	7.2		3.7	11.3	23.9	7.5	6.8	5.4
Often		5.0	8.7		3.8	11.0		9.1	7.3		16.9	7.3	20.2	15.3	12.1	6.5	3.7	13.8		5.5	21.2	29.4	9.5	11.5	6.9
Occasionally		11.7	26.9		14.4	36.5		20.2	24.4		15.6	21.0	31.1	29.2	26.2	13.8	12.7	30.5		12.8	31.8	32.0	21.8	22.9	8.0
Rarely		11.1	26.1		17.1	23.9		21.1	24.7		33.6	24.2	19.8	19.0	19.4	20.2	27.5	16.2		20.9	19.3	12.3	22.4	21.0	5.3
Never*		69.1	34.7		62.8	26.1		41.7	39.4		24.4	40.6	16.2	28.5	37.0	57.2	54.4	32.2		57.0	16.4	2.5	38.8	37.7	17.6
Country of Birth																									
Local*	88.7	88.3	97.4	95.7	97.4	89.5	90.4	93.1	97.8	88.4	97.6	86.2	93.1	94.2	87.3	94.8	92.1	99.8	96.3	87.6	91.0	98.6	89.1	92.8	4.2
Other EU	2.9	5.4	1.7	0.0	0.8	0.0	2.4	6.4	1.2	3.7	0.3	9.8	1.4	0.4	0.0	1.2	3.4	0.0	1.4	4.5	0.0	1.0	0.5	2.1	2.5
Others	8.4	6.4	0.9	4.3	1.8	10.5	7.2	0.5	1.0	7.9	2.0	3.9	5.5	5.5	12.7	4.0	4.4	0.1	2.4	8.0	9.0	0.4	10.5	5.1	3.7

(1) We restrict the sample to households' head aged 26 to 50. We exclude outliers and observations with missing data or showing negative or zero values of income.

(2) The equivalence scale is: e = 1+0.5(N₁₄-1)+0.3N₁₃₋, where N₁₄₊ and N₁₃₋ are the number of household members 14 years of age or older and 13 years of age or younger, repectively.

(3) Data for mother's education in Austria is incomplete (percentages do not add up to one).

Codes: AT- Austria; BE- Belgium; CZ- Czech Republic; DE- Germany; DK- Denmark; EE- Estonia; EL- Greece; FI- Finland; IE- Ireland; ES- Spain; FR- France; IT- Italy; LV- Latvia; LT- Lithuania; HU- Hungary;

NL- The Netherlands; NO- Norway; PL- Poland; PT- Portugal; SI- Slovenia; SK- Slovakia; SE- Sweden; UK- The United Kingdom

*: When data are available, these are the omitted categories in the OLS regression (5). If data are non-available, the omitted category is the next superior.

Variables	AT	BE	CZ	DE	DK	EE	EL
Primary education (F)		-0.021 (0.060)					0.105** (0.054)
Secondary education (F)		0.012		0.175***		-0.034	0.231***
Tertiary education (F)	-0.069	0.028	0.077*	0.126**	0.015	0.131*	0.086
Primary education (M)	(0.100)	0.059	(0.031)	(0.003)	(0.047)	(0.085)	(0.104) 0.084** (0.051)
Secondary education (M)		0.137***		0.192^{***}		0.195***	0.147**
Tertiary education (M)	-0.016	0.138**	0.100^{**}	(0.047) 0.152^{***}	-0.070	(0.00) 0.331^{***} (0.078)	0.365***
Manager (F)	(0.058) 0.114^{**} (0.059)	(0.004) 0.048 (0.068)	0.186**	0.139***	(0.041) 0.068 (0.059)	(0.078) 0.216^{**} (0.124)	0.169***
Proffesional (F)	(0.055) 0.219** (0.118)	(0.000) 0.025 (0.070)	0.227***	(0.040) 0.140^{***} (0.042)	(0.057) 0.030 (0.067)	(0.124) 0.164 (0.131)	0.232**
Technician (F)	(0.110) 0.129^{***} (0.043)	-0.028	0.140**	(0.012) 0.052 (0.041)	(0.060)	0.238**	0.302***
Clerk (F)	(0.013) 0.111** (0.053)	0.013	0.246***	(0.041) (0.026)	(0.000) -0.013 (0.075)	(0.133) 0.304* (0.225)	0.172**
Salesman (F)	0.029	0.017 (0.073)	0.054 (0.083)	0.108**	0.031	0.344**	0.003
Craft trade worker (F)	0.015 (0.036)	-0.022 (0.060)	0.068 (0.061)	0.005 (0.037)	-0.058 (0.047)	0.111 (0.113)	0.124*** (0.051)
Machine operator (F)	-0.025 (0.049)	-0.030 (0.067)	0.066 (0.064)	0.014 (0.041)	-0.050 (0.060)	-0.002 (0.113)	0.064 (0.069)
Elementary occupation (F)	-0.087** (0.041)	-0.039 (0.065)	-0.115 (0.076)	0.077*	-0.008 (0.055)	-0.050 (0.126)	0.005 (0.062)
Armed occupation (F)	0.528*	0.035 (0.093)	0.107 (0.130)	0.077 (0.083)	0.073 (0.157)	-0.132 (0.190)	0.153 (0.181)
Difficulties most of the time		-0.381*** (0.066)	-0.210*** (0.066)		-0.032 (0.098)	-0.158 (0.126)	X
Difficulties often		-0.147*** (0.053)	-0.079** (0.045)		0.031 (0.071)	-0.123** (0.067)	
Difficulties occasionally		-0.139*** (0.036)	-0.034 (0.031)		0.040 (0.040)	0.032 (0.048)	
Difficulties rarely		-0.082** (0.037)	0.001 (0.031)		0.022 (0.037)	0.073* (0.052)	
EU	0.026 (0.067)	-0.084** (0.050)	0.053 (0.091)		0.178 (0.151)		0.132 (0.113)
Other	-0.286***	-0.348*** (0.049)	-0.310***	-0.110*** (0.040)	-0.090 (0.102)	-0.056 (0.061)	-0.495***
Constant	9.765*** (0.029)	9.754*** (0.067)	8.336*** (0.061)	9.402*** (0.068)	(0.049*** (0.040)	(0.124)	8.934*** (0.039)
Observations B sequenced	2155	1838	1589	4255	1241	1377	2126
N-squareu	0.05	0.11	0.00	0.02	0.01	0.09	0.07

 Table 2. Reduced-form OLS regression of household income on circumstances

Standard errors in parenthesis. * significant at 10 %; ** significant at 5 %; *** significant at 1 %.

Omitted categories are: less than primary education; skill agricultural, forestry and fishery worker; never; local. Codes: AT- Austria; BE- Belgium; CZ- Czech Republic; DE- Germany; DK- Denmark; EE- Estonia; EL- Greece; M: Mother. F: Father.

Variables	ES	FI	FR	HU	IE	IT	LT
Primary education (F)	0.178***		0.072**			0.186***	0.371***
	(0.039)		(0.041)			(0.032)	(0.105)
Secondary education (F)	0.234***		0.091**	0.024	0.131***	0.227***	0.377***
	(0.053)		(0.043)	(0.039)	(0.043)	(0.037)	(0.110)
Tertiary education (F)	0.254***	0.086**	0.152***	0.047	0.061	0.371***	0.511***
	(0.064)	(0.038)	(0.052)	(0.057)	(0.059)	(0.068)	(0.126)
Primary education (M)	0.155***		0.078**			0.127***	-0.023
~	(0.037)		(0.041)			(0.028)	(0.096)
Secondary education (M)	0.237***	0.114*	0.142***	0.118***	0.179***	0.191***	0.012
	(0.053)	(0.074)	(0.044)	(0.035)	(0.040)	(0.034)	(0.101)
Tertiary education (M)	0.302***	0.142	0.205***	0.200***	0.149***	0.275***	0.256**
	(0.070)	(0.079)	(0.051)	(0.049)	(0.058)	(0.080)	(0.110)
Manager (F)	0.108**	0.164***	0.202***	0.354***	0.061*	0.075**	0.288**
	(0.055)	(0.048)	(0.034)	(0.055)	(0.042)	(0.038)	(0.126)
Proffesional (F)	0.256***	0.067	0.150***	0.389***	0.197***	0.087*	0.127
	(0.080)	(0.054)	(0.037)	(0.061)	(0.064)	(0.063)	(0.124)
Technician (F)	0.329***	0.047	0.191***	0.272***	0.148**	0.098***	0.165
	(0.063)	(0.041)	(0.034)	(0.053)	(0.087)	(0.042)	(0.149)
Clerk (F)	0.210***	0.049	0.114***	0.202***	0.112**	0.055	0.396***
	(0.058)	(0.083)	(0.038)	(0.062)	(0.064)	(0.045)	(0.161)
Salesman (F)	0.107**	0.079*	0.057	0.164***	0.122**	-0.023	0.401***
	(0.051)	(0.060)	(0.045)	(0.062)	(0.067)	(0.049)	(0.169)
Craft trade worker (F)	0.055*	0.025	0.057**	0.143***		0.001	0.153*
	(0.038)	(0.032)	(0.026)	(0.035)	0.000	(0.030)	(0.097)
Machine operator (F)	0.169***	0.040	0.049**	0.096***	-0.009	0.102***	0.015
	(0.045)	(0.036)	(0.027)	(0.039)	(0.054)	(0.034)	(0.099)
Elementary occupation (F)	0.043	-0.021	-0.007	-0.05/*	-0.053	-0.138***	-0.035
A	(0.040)	(0.063)	(0.034)	(0.042)	(0.046)	(0.034)	(0.099)
Armed occupation (F)	0.228**	0.220**	0.148***	$0.1/2^{**}$	-0.031	0.182***	0.094
	(0.092)	(0.132)	(0.046)	(0.081)	(0.105)	(0.067)	(0.243)
Difficulties most of the time	-0.088**	0.006		-0.104***	-0.2/1***	-0.195***	-0.150**
Difficulting often	(0.045)	(0.056)		(0.038)	(0.059)	(0.033)	(0.083)
Difficulties often	-0.098^{++}	-0.051		-0.080^{++++}	-0.292	-0.103^{++++}	-0.042
Difficulting opposionally	(0.042)	(0.043)		(0.031)	(0.038)	(0.050)	(0.000)
Difficulties occasionally	(0.031)	(0.017)		-0.013	(0.030)	(0.027)	(0.054)
Difficulties revely	(0.051)	0.028)		(0.031)	(0.039) 0.134***	(0.027)	(0.034)
Difficulties farely	(0.031)	-0.003		-0.020	(0.037)	(0.020)	0.013
FU	(0.030)	(0.028)	0.021	(0.020)	(0.037) 0.140	(0.029)	(0.000)
EU	(0.046)	(0.101)	(0.021)	(0.174)	(0.048)	(0.073)	(0.340)
Other	(0.040)	0.218**	0.238***	0.068	0.254***	0.270***	0.040)
Util	(0.150)	(0.100)	(0.031)	-0.008	(0.073)	(0.037)	(0.080)
Constant	8 960***	0.107)	Q 470***	(0.000) 7 Q17***	0.073)	0.037)	7 139***
Constalle	(0.041)	(0.077)	(0.044)	(0.041)	(0.040)	(0.040)	(0.143)
	(0.071)	(0.077)	(0.077)	(0.0-1)	(0.0+0)	(0.0-0)	(0.173)
Observations	5389	1980	3725	2590	1449	8638	1702
R-squared	0.08	0.03	0.09	0.12	0.15	0.07	0.10

Table 2. Reduced-form OLS regression of household income on circumstances (Cont.)

Standard errors in parenthesis. * significant at 10 %; ** significant at 5 %; *** significant at 1 %. Omitted categories are: less than primary education; skill agricultural, forestry and fishery worker; never; local. Codes: FI- Finland; IE- Ireland; ES- Spain; FR- France; IT- Italy; LV- Latvia; LT- Lithuania; M: Mother. F: Father.

Variables	LV	NL	NO	PL	РТ	SE
Primary education (F)				0.067	0.219***	
•				(0.055)	(0.041)	
Secondary education (F)	0.229**	0.069**		0.087*	0.365***	-0.057
	(0.108)	(0.030)		(0.058)	(0.086)	(0.050)
Tertiary education (F)	0.417***	0.162***	-0.020	0.094	0.728***	0.083*
	(0.146)	(0.041)	(0.040)	(0.086)	(0.149)	(0.064)
Primary education (M)				0.057	0.116***	-0.100
				(0.052)	(0.038)	(0.140)
Secondary education (M)	0.113	0.017		0.210***	0.179**	-0.015
	(0.105)	(0.028)		(0.054)	(0.101)	(0.145)
Tertiary education (M)	0.201*	-0.021	0.031	0.344***	0.308***	-0.060
	(0.130)	(0.046)	(0.040)	(0.072)	(0.111)	(0.150)
Manager (F)	0.242	0.011	0.010	0.257***	0.385***	
	(0.210)	(0.033)	(0.072)	(0.064)	(0.068)	
Proffesional (F)	0.271*	-0.013	0.015	0.439***	0.256*	
	(0.209)	(0.043)	(0.079)	(0.077)	(0.159)	
Technician (F)	0.277*	0.074**	-0.001	0.257***	0.446***	
	(0.211)	(0.037)	(0.068)	(0.050)	(0.092)	
Clerk (F)	-0.290	0.088**	0.152*	0.204***	0.287***	
	(0.303)	(0.046)	(0.095)	(0.063)	(0.079)	
Salesman (F)	0.240	-0.095**	-0.018	0.085	0.331***	
	(0.255)	(0.055)	(0.094)	(0.072)	(0.073)	
Craft trade worker (F)	0.104	. ,	-0.038	0.132***	0.125***	
	(0.180)		(0.064)	(0.031)	(0.043)	
Machine operator (F)	0.181	0.004	0.038	0.140***	0.100**	
-	(0.179)	(0.042)	(0.070)	(0.034)	(0.055)	
Elementary occupation (F)	0.113	-0.022	0.003	0.031	0.127**	
• • •	(0.191)	(0.058)	(0.177)	(0.041)	(0.056)	
Armed occupation (F)	0.156	-0.032	-0.076	0.350***	0.534***	
	(0.273)	(0.095)	(0.183)	(0.089)	(0.139)	
Difficulties most of the time	-0.023	-0.154**	-0.045	-0.240***	. ,	0.129
	(0.136)	(0.072)	(0.129)	(0.043)		(0.107)
Difficulties often	0.006	-0.057*	-0.033	-0.162***		0.037
	(0.096)	(0.045)	(0.091)	(0.034)		(0.088)
Difficulties occasionally	-0.043	-0.056**	-0.105**	-0.056**		0.030
-	(0.073)	(0.032)	(0.053)	(0.027)		(0.061)
Difficulties rarely	0.022	-0.021	-0.056*	-0.046*		0.021
ě	(0.080)	(0.028)	(0.039)	(0.031)		(0.050)
EU	~ /	0.107	0.105	× ,	-0.173*	-0.155*
		(0.099)	(0.092)		(0.129)	(0.096)
Other	-0.127*	-0.213***	-0.372***	-0.369*	-0.147*	-0.473***
	(0.088)	(0.055)	(0.083)	(0.282)	(0.101)	(0.083)
Constant	7.235***	9.734***	10.178***	7.543***	8.543***	9.828***
	(0.201)	(0.034)	(0.058)	(0.040)	(0.036)	(0.141)
		、 /			< - /	. /
Observations	1159	1695	1423	6055	1654	1342
R-squared	0.05	0.04	0.03	0.08	0.20	0.04

Table 2. Reduced-form OLS regression of household income on circumstances (Cont.)

Standard errors in parenthesis. * significant at 10 %; ** significant at 5 %; *** significant at 1 %.

Omitted categories are: less than primary education; skill agricultural, forestry and fishery worker; never; local.

Codes: HU-Hungary; NL- The Netherlands; NO- Norway; PL- Poland; PT- Portugal; SI- Slovenia;

M: Mother. F: Father.

Variables	SI	SK	UK
Drimony advantion (E)	0 17(**		
Frinary education (F)	(0.087)		
Secondary education (F)	(0.087) 0.246***	0.006	0 1/0***
Secondary education (1)	(0.088)	(0.073)	(0.051)
Tertiary education (F)	0.210**	0.076	0 191***
for dary education (1)	(0.109)	(0.086)	(0.046)
Primary education (M)	-0.126**	(0.000)	(0.0.0)
	(0.074)		
Secondary education (M)	-0.062	0.089	0.171***
v ()	(0.077)	(0.070)	(0.044)
Tertiary education (M)	-0.064	0.184**	0.122***
•	(0.096)	(0.085)	(0.049)
Manager (F)	0.284***	0.151**	0.247***
	(0.077)	(0.075)	(0.103)
Proffesional (F)	0.127*	0.188***	0.185**
	(0.087)	(0.079)	(0.104)
Technician (F)	0.173***	0.178***	0.112
	(0.057)	(0.071)	(0.108)
Clerk (F)	0.088*	0.193**	0.007
	(0.065)	(0.088)	(0.123)
Salesman (F)	0.115**	0.144*	0.179*
	(0.063)	(0.088)	(0.122)
Craft trade worker (F)	0.008	0.103*	0.083
	(0.044)	(0.066)	(0.098)
Machine operator (F)	0.029	0.080	0.085
	(0.043)	(0.067)	(0.100)
Elementary occupation (F)	0.009	0.001	0.043
	(0.058)	(0.069)	(0.102)
Armed occupation (F)	0.243**		
	(0.134)		
Difficulties most of the time	0.010	-0.006	-0.105*
	(0.049)	(0.070)	(0.064)
Difficulties often	0.026	-0.017	-0.003
	(0.042)	(0.069)	(0.058)
Difficulties occasionally	0.024	-0.054	0.072**
	(0.038)	(0.069)	(0.043)
Difficulties rarely	0.067	-0.062	-0.043
	(0.040)	(0.073)	(0.042)
EU		0.165*	-0.005
04	0 170***	(0.105)	(0.228)
Other	-0.1/9***	-0.138	-0.225***
Constant	(0.043)	(0.158)	(0.052)
Constant	8.942***	(0.101)	9./51***
	(0.082)	(0.101)	(0.096)
Observations	1202	2203	1974
R-squared	0.08	0.04	0.08
ix oquat cu	0.00	0.01	0.00

Table 2. Reduced-form OLS regression of household income on circumstances (Cont.)

Standard errors in parenthesis. * significant at 10 %; ** significant at 5 %; *** significant at 1 %.

Omitted categories are: less than primary education; skill agricultural, forestry and fishery worker; never; local.

United Kingdom: occupation variables are refered to mother's occupation.

Codes: SK- Slovakia; SE- Sweden; UK- The United Kingdom

M: Mother. F: Father.

Table 3. Indices of total inequality	, absolute IO and relat	tive IO in Europe (2005).
--------------------------------------	-------------------------	---------------------------

Index	AT	BE	CZ	DE	DK	EE	EL	ES	FI	FR	HU	IE	IT	LT	LV	NL	NO	PL	РТ	SE	SI	SK	UK
The 10	0 1101	0 1021	0.1106	0 1205	0.0690	0.1095	0.2127	0.2144	0.1160	0.1006	0.1214	0.1611	0 1074	0.2492	0 2005	0.0004	0.1160	0.2640	0.2264	0 1005	0.0972	0 1251	0 1052
I nell 0	(0.0062)	(0.0075)	(0.0079)	(0.0052)	(0.0083)	(0.0115)	(0.0126)	(0.2144) (0.0087)	(0.0066)	(0.0035)	(0.1314)	(0.0084)	(0.0065)	(0.0144)	(0.0242)	(0.0052)	(0.0118)	(0.0078)	(0.0113)	(0.0152)	(0.0059)	(0.0068)	(0.0115)
ю	0.0060	0.0123	0.0070	0.0027	0.0013	0.0218	0.0230	0.0286	0.0038	0.0097	0.0152	0.0242	0.0220	0.0358	0.0213	0.0041	0.0048	0.0272	0.0503	0.0087	0.0084	0.0045	0.0199
	(0.0012)	(0.0032)	(0.0016)	(0.0006)	(0.0009)	(0.0047)	(0.0034)	(0.0028)	(0.0011)	(0.0011)	(0.0018)	(0.0031)	(0.0023)	(0.0065)	(0.0078)	(0.0011)	(0.0033)	(0.0030)	(0.0061)	(0.0054)	(0.0016)	(0.0013)	(0.0034)
Ratio (%)	5.08	11.93	5.85	2.07	1.89	10.98	10.81	13.34	3.28	8.85	11.57	15.02	11.74	14.42	7.11	4.64	4.11	10.27	22.22	7.95	9.62	3.60	10.19
	(0.97)	(2.44)	(1.26)	(0.46)	(0.96)	(1.95)	(1.55)	(1.13)	(0.88)	(1.00)	(1.35)	(1.88)	(1.05)	(2.15)	(2.45)	(1.13)	(2.48)	(0.99)	(2.22)	(3.81)	(1.85)	(0.96)	(1.61)
T0 position	9	4	10	12	1	17	18	19	7	6	13	14	15	21	23	3	8	22	20	5	2	11	16
IO position	7	12	8	2	1	16	18	21	3	11	13	19	17	22	15	4	6	20	23	10	9	5	14
N	2155	1838	1589	4255	1241	1377	2126	5389	1980	3725	2590	1449	8638	1702	1159	1695	1423	6055	1654	1342	1393	2292	1874

Table 4. Correlation of overall inequality and IO with alternative indicators.

	Lag corr (1998 V	elation * s. 2005)	Contemporaneous correlation* (2005 Vs. 2005)			
Development indicators	Theil 0	10	Theil 0	10		
GDP	-0.6240	-0.4679	-0.5772	-0.4421		
% empl. Agric.	0.7287	0.5754	0.6761	0.6289		
% empl. Serv.	-0.5135	-0.4653	-0.4812	-0.4758		
Education indicators	Theil 0	ю	Theil 0	ю		
Second.attained (total)	-0.2072	-0.5855	-0.2177	-0.5972		
Second.attained (female)	-0.1493	-0.5228	-0.1515	-0.5333		
Tertiary-upper attained (total)	-0.0032	0.0772	0.3697	0.2280		
Early leaves	0.5333	0.7852	0.3595	0.6620		
Labor market indicators	Theil 0	ю	Theil 0	ю		
Employment rate	-0.3585	-0.3007	-0.4020	-0.2678		
Employment rate (female)	-0.3610	-0.3810	-0.4248	-0.3638		
Unemployment rate	0.5234	0.2790	0.3596	0.1311		
Long-run unemployment	0.4485	0.4604	0.2748	0.2142		
Social public expenditures items	Theil 0	IO	Theil 0	ю		
Total	-0.6746	-0.6049	-0.6222	-0.4181		
Social protection, total	-0.6689	-0.6403	-0.6245	-0.4346		
Unemployment	-0.5649	-0.4365	-0.4531	-0.2750		
Old persons	-0.2115	-0.3355	-0.3104	-0.2079		
Health	-0.7539	-0.5517	-0.6587	-0.3617		
Social exclusion	-0.6111	-0.5422	-0.6271	-0.5624		
Disability	-0.4391	-0.4184	-0.5061	-0.4063		
Child care	-0.6507	-0.6994	-0.6827	-0.6565		

(*): Lag correlation: Corr.(Ineq.1998, Indicator2005); Contemporaneous correlation: Corr.(Ineq.2005, Indicator2005)

Figure 1. Total inequality in Europe (2005). (Theil 0 index)



Figure 2. Absolute inequality of opportunity in Europe (2005). (Theil 0 index)





Figure 3. Relative inequality of opportunity in Europe (2005). (Theil 0 index)

Figure 4. Development and IO in Europe.





Figure 5. Long-run unemployment rate and IO in Europe.

Figure 6. Early school-leavers and IO in Europe.





Figure 7. Public expenditure in social protection and IO in Europe.

APPENDIX

	Deve	lopment indic	ators	Labor Market indicator						
	GDP per capita (PPP adjusted)	Employ. in agriculture	Employ. in service	Employ. rate	Female employ. rate	Unempl. rate	Long-term unempl. (12 months or more)			
Year	2005	2005	2005	2005	2005	2005	2005			
Source	Eurostat,	Eurostat,	Eurostat,	Eurostat,	Eurostat,	Eurostat,	Eurostat,			
	National	Labour Force	Labour Force	Labour Force	Labour Force	Labour Force	Labour Force			
	accounts	Survey	Survey	Survey	Survey	Survey	Survey			
Units	Millions of	% over total	% over total	% population	%	% active	%unemploy.			
	PPP/habitant			15-64	population,	population				
					fem. 15-64					
AT	0.0279	6.0	70.1	75.0	62.0	5.2	25.3			
BE	0.0268	2.0	77.5	65.9	53.8	8.5	51.7			
CZ	0.0171	3.8	57.9	71.7	56.3	7.9	53.0			
DE	0.0263	2.2	71.9	75.5	60.6	10.7	53.0			
DK	0.0278	3.0	76.1	82.3	71.9	4.8	23.4			
EE	0.0139	5.3	61.0	69.1	62.1	7.9	53.4			
EL	0.0206	12.4	67.7	64.2	46.1	9.9	52.2			
ES	0.0229	5.2	65.5	68.9	51.2	9.2	24.5			
FI	0.0257	5.1	69.1	72.8	66.5	8.4	25.8			
FR	0.0249	3.6	76.0	66.0	58.5	9.3	41.1			
HU	0.0142	8.7	59.7	63.3	51.0	7.2	45.0			
IE	0.0323	5.9	66.5	71.8	58.3	4.4	33.4			
IT	0.0236	4.2	67.0	68.0	45.3	7.7	49.9			
LT	0.0119	14.0	57.1	65.7	59.4	8.3	52.5			
LV	0.0109	11.2	62.3	67.9	59.3	8.9	46.0			
NL	0.0294	3.2	79.3	79.1	66.4	4.7	40.2			
NO	0.0396	3.3	76.7	83.4	71.7	4.5	18.7			
PL	0.0115	17.4	53.5	54.5	46.8	17.8	57.7			
РТ	0.0173	11.9	58.8	76.2	61.7	7.7	48.2			
SE	0.0271	2.3	74.8	80.3	70.4	7.4	18.0			
SI	0.0197	10.0	54.9	68.5	61.3	6.5	47.3			
SK	0.0135	4.4	61.6	56.0	50.9	16.3	71.9			
UK	0.0274	1.5	80.3	77.5	65.8	4.8	21.1			

Table 1A. Descriptive statistics for the independent variables

Table 1A.	Descriptive	statistics fo	r the indepe	endent var	iables (Cont.)
	Descriptive	statistics io	i une maep	chucht van	

		Education	indicators		Expenditure in Social protection and taxes								
	Educ. attained (at least secondary)	Educ. female attained (at least secondary)	Early school- leavers	Educ. attained, at least Tertiary	Total expenditure in social protection	Health care/Sickness	Disability	Child care	Social exclusion	Unemploy. benefits	Old age		
Year Source	2005 Eurostat (ISCED)	2005 Eurostat (ISCED)	2005 Eurostat (ISCED)	2005 Eurostat (ISCED)	2005 Eurostat, Living conditions and welfare								
Units	% population 25-64	% population fem. 25-64	% people 18- 24, at most lower second. educ.	%, population 24-65	% GDP								
AT	80.6	75.4	9.00	4.7	23.7	7.1	2.4	1.2	0.2	1.4	9.8		
BE	66.1	65.9	13.00	6.1	25.9	6.7	3.3	3.0	0.5	2.4	8.7		
CZ	89.9	86.3	6.40	4.8	21.5	6.4	2.1	2.5	0.1	0.6	8.8		
DE	83.1	79.7	13.80	4.4	25.8	8.0	2.3	1.6	0.2	0.7	10.8		
DK	81.0	79.6	8.50	6.9	29.3	6.1	4.2	3.8	1.0	2.5	11.0		
EE	89.1	90.9	14.00	7.8	28.3	7.7	2.0	2.0	0.4	3.5	9.8		
EL	60.0	58.9	13.30	9.1	19.2	3.8	2.0	0.8	0.4	0.6	9.3		
ES	48.5	48.5	30.80	6.5	23.6	6.6	1.2	1.5	0.5	1.2	11.3		
FI	78.8	81.0	9.30	9.3	28.5	8.1	1.8	3.3	0.2	2.0	10.3		
FR	66.8	64.9	12.00	5.5	23.4	7.5	4.5	2.8	0.7	0.6	6.9		
HU	76.4	72.7	12.30	6.6	30.9	8.0	4.6	2.9	0.6	1.9	11.7		
IE	65.2	68.1	12.30	6.8	12.5	4.0	1.2	1.5	0.1	0.2	5.4		
IT	50.4	50.6	21.90	5.6	20.6	6.3	1.5	1.2	0.2	2.5	8.0		
LT	87.6	88.6	9.20	8.8	12.7	3.9	1.3	1.2	0.2	0.2	5.5		
LV	84.5	87.4	11.90	8.7	25.4	6.8	1.5	1.1	0.0	0.5	12.9		
NL	71.8	68.4	13.60	5.4	11.8	3.1	0.9	1.3	0.1	0.5	5.6		
NO	88.2	88.0	4.60	7.6	16.1	4.8	1.3	1.9	0.5	0.5	6.9		
PL	84.8	83.9	5.50	8.2	22.5	7.3	1.9	1.9	0.6	0.7	9.5		
PT	26.5	28.6	38.60	5.7	26.0	8.0	2.5	1.3	1.2	1.6	9.7		
SE	83.6	85.7	11.70	7.9	18.5	6.5	1.4	1.4	0.5	0.7	7.1		
SI	80.3	77.4	4.30	8.4	29.5	8.8	1.7	2.5	0.5	2.2	11.0		
SK	87.9	84.7	5.80	4.9	27.9	7.1	2.4	3.0	0.3	1.6	11.3		
UK	71.8	67.5	14.00	6.2	16.9	6.9	0.9	2.5	0.3	1.3	3.8		